Author's response to reviews

Title: Primary care Identification and Referral to Improve Safety of women experiencing domestic violence (IRIS): protocol for a pragmatic cluster randomised controlled trial

Authors:

Alison Gregory (alison.gregory@bristol.ac.uk)
Jean Ramsay (j.m.ramsay@qmul.ac.uk)
Roxane Agnew-Davies (RoxaneDavies@aol.com)
Kathleen Baird (kathleen2.baird@uwe.ac.uk)
Angela Devine (a.devine@qmul.ac.uk)
Danielle Dunne (d.e.dunne@qmul.ac.uk)
Sandra Eldridge (s.eldridge@qmul.ac.uk)
Annie Howell (AHowell@niaproject.plus.com)
Medina Johnson (medina.johnson@nextlink.housing.co.uk)
Clare Rutterford (c.m.rutterford@qmul.ac.uk)
Debbie Sharp (debbie.sharp@bristol.ac.uk)
Gene Feder (gene.feder@bristol.ac.uk)

Version: 3 Date: 27 January 2010

Author's response to reviews: see over
January 27th 2010

Editorial board
BMC Public Health

Dear colleagues

Primary care Identification and Referral to Improve Safety of women experiencing domestic violence (IRIS): protocol for a pragmatic cluster randomised controlled trial

In response to your e-mail, we have edited our manuscript to address the major and minor revisions mentioned. We have made formatting changes to the list of authors so they reflect what is written in the submission system, and we have added a statement to indicate that all of our authors have read and approved the final manuscript. We have altered the formatting of our references in the body of the text and the layout of the front sheet. The only other change is the addition of a twelfth author who we had previously omitted in error.

Yours sincerely

[Signature]